Author's response to reviews

Title: Outbreak of viral hemorrhagic fever caused by dengue virus type 3 in Al-Mukalla, Yemen

Authors:

Tariq A Madani (tmadani@kau.edu.sa)  
El-Tayeb ME Abuelzein (eabuelzein@yahoo.com)  
Hussein MS Al-Bar (husalbar@hotmail.com)  
Esam I Azhar (eazhar@kau.edu.sa)  
Moujahed Kao (moujahedkao2005@yahoo.de)  
Haj O Alshoeb (alshoeb@hotmail.com)  
Alabd R Bamoosa (d.bamosa@hotmail.com)

Version: 3 Date: 14 February 2013

Author's response to reviews:

6 February 2013

The Editor, BMC Infectious Diseases

Re: Manuscript ID: 1698722919783709

Article Title: Outbreak of viral hemorrhagic fever caused by dengue virus type 3 in Al- Mukalla, Yemen.

Dear the Editor

Thank you very much for accepting the above mentioned manuscript. As per your request, we have revised the Ethical approval section of the manuscript to state that ethical approval for the animal experiments was obtained from the Research Ethics Committee of the Faculty of Medicine, King Abdulaziz University, Jeddah, Saudi Arabia. This sections now reads as follows:

Ethical approval

Ethical approval for the human and animal research and experiments was obtained from the Research Ethics Committee of the Faculty of Medicine, King Abdulaziz University, Jeddah, Saudi Arabia.

Thank you

Corresponding author

Tariq A. Madani